Author's response to reviews

Title: Differential diagnosis of tuberculous meningitis from partially-treated pyogenic meningitis by cell ELISA.

Authors:

Rajpal SINGH Kashyap (raj_ciims@redifmail.com)
Rani POONAM. Kainthla (poonam_ciims@yahoo.com)
Ravindra MANIKRAO Satpute (ravi_ciims@yahoo.co.in)
Neha PANKAJ Agarwal (neha_ciims@yahoo.co.in)
Nitin H Chandak (ciims_ngp@sanchar.net.in)
Hemant JYOTISWAROOP Purohit (hemanttdrd@rediffmail.com)
Girdhar MADANGOPAL Taori (taorigm_ciims@yahoo.com)
Hatim FATEH Daginawala (hfd_ciims@rediffmail.com)

Version: 8 Date: 20 October 2004

Author's response to reviews:

To,
The Editor
BMC Neurology,

Please find enclosed the final version of the manuscript entitled "Differential diagnosis of tuberculous meningitis from partially-treated pyogenic meningitis by cell ELISA" along with the formatting changes as suggested by the Editorial Administrator.

With Kind Regards
Dr.H.F.Daginawala